Author's response to reviews

**Title:** Cognitive behavioural therapy versus supportive therapy for persistent positive symptoms in psychotic disorders: The POSITIVE Study, a multicenter, prospective, single-blind, randomised controlled clinical trial

**Authors:**

- Stefan Klingberg (stefan.klingberg@med.uni-tuebingen.de)
- Andreas Wittorf (andreas.wittorf@med.uni-tuebingen.de)
- Christoph Meisner (christoph.meisner@med.uni-tuebingen.de)
- Wolfgang Wölwer (wolfgang.woelwer@uni-duesseldorf.de)
- Georg Wiedemann (georg.wiedemann@klinikum-fulda.de)
- Jutta Herrlich (jutta.herrlich@kgu.de)
- Andreas Bechdolf (andreas.bechdolf@uk-koeln.de)
- Bernhard W Mueller (bernhard.mueller@uni-due.de)
- Gudrun Sartory (sartory@uni-wuppertal.de)
- Michael Wagner (michael.wagner@ukb.uni-bonn.de)
- Tilo Kircher (kircher@med.uni-marburg.de)
- Hans-Helmut Koenig (h.koenig@uke.uni-hamburg.de)
- Corinna Engel (corinna.engel@med.uni-tuebingen.de)
- Gerhard Buchkremer (helma.buxmann@med.uni-tuebingen.de)

**Version:** 6  **Date:** 7 December 2010

**Author's response to reviews:** see over
Re-Resubmission to “Trials”: Cognitive behavioural therapy versus supportive therapy for persistent positive symptoms in psychotic disorders: The POSITIVE Study, a multicenter, prospective, single-blind, randomised controlled clinical trial

Dear Editors:

Thank you for your email of 3rd December and the information that this manuscript is now in the proofing stage. We have carefully revised the manuscript according to your requests.

- We replaced reference 26 by an published article (Flückiger et al., 2010)
- References already had been in numerical order. There are some references which are first cited in the introduction and again later in the manuscript (e.g. citation number 7 “Tarrier et al.” is cited on page 5 and 29). These later citations may have led to the misapprehension that references were not in numerical order as these citations keep their number (e.g. 7) throughout the manuscript.
- We finally checked typography.

Other changes have not been made.

We now think that the manuscript is ready for publication. If you have any further requests, please let us know.

Thank you for accepting this paper for publication

Sincerely

Prof. Dr. Stefan Klingberg